Author's response to reviews

Title: Amsterdam Wrist Rules, A Clinical Decision Aid

Authors:

Abdel Bentohami MD (a.bentohami@amc.nl)
Monique M.J. Walenkamp Msc (m.m.walenkamp@amc.nl)
Suzan Beerekamp MD (m.s.beerekamp@amc.nl)
Annelie Slaar MD (a.slaar@amc.nl)
Eva M. Verhoog MD (e.m.verhoog@amc.nl)
Bart A. van Dijkman MD (bvdijkman@flevoziekenhuis.nl)
Taco S. Bijlsma MD, PhD (bijlsmat@spaarneziekenhuis.nl)
Marco Maas MD, PhD (m.maas@amc.nl)
Cara Jager MD (l.c.jager@amc.nl)
Niels W.L. Schep MD, Msc, PhD (n.w.schep@amc.nl)
Carel Goslings MD, PhD, Prof. (j.c.goslings@amc.nl)

Version: 2 Date: 14 July 2011

Author's response to reviews: see over
Dear editors,

Thank you very much for your response regarding our manuscript; *The Amsterdam Wrist Rules, a clinical decision aid*.

Attached to this email you will find a point by point response for both reviewers in separate documents, the revised manuscript with changes tracked and the clean manuscript in which these changes have been accepted.

Regarding the editorial comment on ethical approval, the authors added the required information to page 9, last paragraph.

Hopefully we have managed to successfully answer all questions and concerns regarding our manuscript and thereby improved the quality.

On behalf of all members of the AWR project group,

With kind regards,

M.M.J. Walenkamp, MSc